Author’s response to reviews

Title: Pneumonia due to Pandoraea Apista after Evacuation of Traumatic Intracranial Hematomas: A case report and literature review

Authors:

Chuanzhong Lin (xinsibugu@163.com)

Ning Luo (lnluoning@163.com)

Qiang Xu (xuqiang911@zju.edu.cn)

Jianjun Zhang (szyyxrx@163.com)

Mengting Cai (caimt3@mail2.sysu.edu.cn)

Guanhao Zheng (ken_zheng@foxmail.com)

Ping Yang (yangp2009@126.com)

Version: 2 Date: 26 Aug 2019

Author’s response to reviews:

Dear Dr. Oliver Cocks,

Thank you very much for your letter and advice. We have revised the manuscript according to your kind advices. We sincerely hope this manuscript will be finally acceptable to be published on BMC Infectious Diseases. Thank you very much for all your help and looking forward to hearing from you soon.

We would like to express our great appreciation to you and reviewers for comments on our paper. Looking forward to hearing from you.

Thank you and best regards.

Yours sincerely,

Ping Yang

List of Responses
Response to editor:

Editor Comments:

1. Please state in the cover letter whether the map depicted in figure 3 is your own or taken from another source.

If taken from another source please acknowledge the source in the figure legend, and if it is under copyright also state the written permission given to use and adapt it.

If the above conditions are not met the image needs to be removed. Please note the editors may request proof of permission at any time.

Should you require an alternative source you may wish to try Wikimedia Commons:

http://commons.wikimedia.org/wiki/Main_Page

If images were taken from multiple sources please clarify whether permission was obtained from all sources to use and alter these images.

Response: Thank you very much for your friendly reminder and kindly suggestion. The original figure 3 was based on a blank world map, which was gotten on the Web. And yet, we couldn’t find the original page and its copyright statement anymore. Therefore, we used R software (Version 3.4.4) to create a new blank world map, and edited it by Microsoft Office PowerPoint 2016. We can guarantee you that the new map depicted in figure 3 is created by our own, and we have all the copyright to use and adapt it.

2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: Thank you very much for your friendly reminder. All the tracked changes, comments, highlights and strikethroughs have been removed from our manuscript. And text in different colours has been revised.

Response to Toshinori Kawanami, M.D., Ph.D. (Reviewer 1):

Reviewer reports:

Toshinori Kawanami, M.D., Ph.D. (Reviewer 1): Thank you for your response to my comments. I believe the paper will be of interest to the readership of BMC Infectious Diseases, and the revised manuscript is acceptable for publication.
Response: Thank you very much for your kindly review. Special thanks to you for your positive comments.

We appreciate for Editors/Reviewers’ warm work earnestly and hope that the correction will meet with approval.

Once again, thank you very much for your comments and suggestions.

Yours sincerely,

Ping Yang